Author’s response to reviews

Title: Effect of a tailored leaflet to promote diabetic retinopathy screening among young adults with type 2 diabetes: a randomised controlled trial.

Authors:

Amelia Lake (alake@acbrd.org.au;alake@deakin.edu.au)

Jessica Browne (jhateleybrowne@ceiglobal.org)

Gwyneth Rees (Gwyneth.rees@gmail.com)

Jane Speight (jspeight@acbrd.org.au)

Version: 2 Date: 03 Jan 2020

Author’s response to reviews:

Please refer to separate 'Letter to the Editor' uploaded with this revision.